EuroIntervention

Chronic total coronary occlusions and the Occluded Artery Trial. A critical appraisal.

Gerald S. Werner^{1*}, MD, FACC, FESC; Carlo Di Mario², MD, FACC, FRCP, FESC; Alfredo R. Galassi³, MD, FACC, FESC; Anthony Gershlick⁴, MD, FRCP, FESC; Bernhard Reimers⁵, MD, FESC; George Sianos⁶, MD, PhD, FESC; Horst Sievert⁷, MD, PhD, FESC; Thierry Lefevre⁸, MD, FESC; Nicolaus Reifart⁹, MD, PhD, FESC for the EuroCTO Club

 Klinikum Darmstadt, Darmstadt, Germany; 2. Royal Brompton Hospital & Imperial College, London, United Kingdom;
Ospedale Ferrarotto, Univ. of Catania, Catania, Italy; 4. Univ. of Leicester, Leicester, United Kingdom; 5. Ospedale di Mirano, Venice, Italy; 6. Thoraxcentre, Erasmus University, Rotterdam, The Netherlands; 7. Katharinenhospital Frankfurt, Frankfurt, Germany; 8. Institut Cardiovasculaire Paris Sud, Massy, Paris, France; 9. Main-Taunus-Klinik Bad Soden, Germany

There exists no conflict of interest for any of the authors.

One of the highlights of the recent AHA 2006 was the long-awaited presentation of the Occluded Artery Trial (OAT)¹, a commendable effort of a large group of investigators headed by Dr. Judith Hochman to address the open artery hypothesis² in a large-scale randomised trial. Within only few months, several editorials and comments in various journals took up the presented data and drew often far-reaching conclusions on the futility of treating occluded coronary arteries in general^{3,4}. This included also chronically occluded arteries (CTOs), which are by definition arteries occluded for more than three months and supplied by collateral arteries⁵. But these lesions were not included in OAT, a study dealing with recent occlusions post-Ml⁶.

The open artery hypothesis was a widely accepted concept despite the absence of confirmation from randomised controlled trials⁷. The question now arises whether the new data presented by the OAT trial indeed replace and forfeit the open artery concept. The present comment and discussion of the OAT trial will address these issues on the basis of a critical analysis of the published data, and discuss strength but also pitfalls and shortcomings of these data. We will also try to better define the patient population for whom the conclusions drawn by the authors and commentators are applicable.

The OAT results

In OAT patients with a recent myocardial infarction (MI) of 3-28 days with angiographically confirmed diagnosis of an occluded infarct-related artery, were randomised to medical treatment or to an interventional attempt to open the occluded infarct-related artery. During a mean follow-up of approximately three years there was no advantage of the interventional approach in terms of survival and there were more recurrent MIs than with the conservative approach¹. These data were further corroborated by a substudy conducted in Canada, the TOSCA-2 trial, which looked at changes of left ventricular function during follow-up, and observed a slight improvement of LV function over time but no difference between the treatment arms⁸. The validity and integrity of these data are not questioned, but there are considerable issues with their generalisation to daily clinical practice. There is no doubt that primary PCL for acute ML is

practice. There is no doubt that primary PCI for acute MI is lifesaving in the short and long-run, proven by several randomised trials with different time windows after symptom onset^{9,10}. Also, in patients with no access to primary PCI within 90 min and therefore treated with thrombolysis, there is concordant evidence from the randomised studies¹¹⁻¹⁴ that the transfer to angioplasty in the first 24 hours is clinically beneficial. There are clear indications to primary PCI or early angiography and angioplasty after thrombolysis

* Corresponding author: Medizinische Klinik I, Klinikum Darmstadt, Grafenstrasse 9, D-64283 Darmstadt, Germany E-mail: gerald.werner@klinikum-darmstadt.de

© Europa Edition 2008. All rights reserved.



in the current guidelines. If the guidelines are properly applied, only a minority of AMI patients are likely to still have an occluded artery after three or more days from the acute episode and then meet the eligibility criteria for OAT.

OAT and subacutely occluded arteries after MI

A total of 2,166 patients finally enrolled during five years in 217 centres which was lower than originally planned, yet still ranks this study among the largest interventional studies. Yet, those randomised represent a small highly selected group of the population actually treated in the participating centres. This important information is essential; studies like the recent COURAGE trial, when less than 6% of patients screened were actually enrolled, are unlikely to have general application for a population of patients with chronic angina¹⁵. The OAT publication lacks important information - that is the number of patients screened and potentially eligible and why they were not enrolled - especially when only one patient per year and centre was enrolled in the US. The low inclusion number per centre in a population which is probably more common in clinical practice suggests that the patients finally randomised were highly selected patients, and that the screening process before randomisation was subjective and not transparent to the reader of the paper. The inclusion criteria should guarantee a high-risk study population, but one of the two major risk criteria, a moderately impaired ejection fraction <50%, was present in only about half of all patients, and a more severely reduced ejection fraction of below 40, was present in just 20%. The overall mortality was 9% within four years, which is surprisingly low and atypical for a post MI patient group. This suggests, that the study group was in fact not a high-risk subset.

This was most probably driven by the tendency of the investigators to include patients rather selectively in the trial. The high prevalence of single-vessel disease (85%) is not typical of an infarct population and demonstrates that patients with advanced disease were not considered for randomisation¹⁶. The initial power analysis expected a 3-year event rate in the medical group of 25%, the actual rate was only 15.6% after four years, which leaves the study statistically underpowered. And in fact, no conclusion should be drawn from a study which is underpowered.

The patient population was defined to represent occluded arteries after an acute MI within four weeks, but the actual time delay was eight days as median. None, or only mild ischaemia in the territory supplied by the occluded artery was present in 90% of the patients enrolled, and this was tested in only one fourth by an actual stress test. How the presence of ischaemia was determined in the other patients is not explained.

Only 20% of patients had undergone thrombolysis, whereas we should expect the standard of care to be PCI or at least thrombolysis for acute MI, and a population like this – with recent MI – should become the exception rather than the rule in developed countries. For acute MI interventions our accepted goal is to achieve TIMI 3 flow which is achieved in about 95% of patients, and it is not acceptable as standard of care that TIMI 2 flow was considered a successful PCI result in OAT. We learned from the thrombolysis era that TIMI 2 flow does not translate into a good clinical outcome¹⁷. In

fact, in almost 20% of patients even TIMI 2 flow had not been achieved, which represents a low interventional success rate, impacting the intention-to-treat based outcome analysis. Furthermore, the long-term patency achieved with PCI was low as reported in the TOSCA-2 substudy with TIMI 3 in only 75% of patients⁸.

One other issue that is not mentioned in the publication, nor in the comments, is the fact that the study reports a 5-year follow-up, but only 45% of the patients had even completed the 3-year follow-up, and just 25% were into the five years of reported follow-up duration. Based on the study hypothesis, a prognostic intervention would require a longer follow-up to show effects if the individual patient risk is as low as in OAT.

OAT and the conclusions from non-significant results

Based on the results achieved in the interventional arm of an obviously low-risk population of asymptomatic post-MI patients, the authors conclude that PCI should not be recommended in this setting. They state that nonfatal MI tended to be increased after PCI (p=0.08) and this was the concluding remark in the abstract, a questionable statement to highlight a trend when a large study of more than 2,000 patients did not yield statistically significant endpoint results, and the prespecified power was not achieved. Furthermore, the PCI success rate, per se, was lower than one would expect in these subacute occlusions, with a considerable number of periprocedural MIs. Both factors would attenuate any possible beneficial effect of PCI in these patients.

The way the results were interpreted, which was not supported by statistical significance, is not acceptable and reveals an underlying bias on the part of the authors. Should we not also accept the fact that, based on OAT, both strategies are equal regarding hard endpoints (death and MI)? Based on the OAT results, the physician can discuss with his patient who had an acute MI on his holiday trip and returned to undergo a diagnostic procedure one week after the event to find a proximally occluded LAD, whether or not to leave this situation as it is, or to reopen it. He should inform his patient that no proven benefit was observed within two to three years of follow-up in a randomised study, but that there is no severe disadvantage to be expected. Furthermore, if we cannot stop atherosclerotic progression in this patient – and who can guarantee this today? – there is the risk of another event in the future with a higher fatality if there is already an occluded artery^{16,18}.

TOSCA-2 and the open artery hypothesis

One key concept of the open artery hypothesis is the improvement of LV function after late reopening of an infarct-related artery. The benefit of such an intervention was not observed in the TOSCA-2 trial, as a moderate improvement of LV function was observed also in the conservatively treated study arm. However, the interventional success in these subacute MI patients was less than what we would consider optimal for PCI in acute MI. In one of the recent acute MI trials, TIMI 3 was achieved in 95% of lesions¹⁹, in TOSCA-2, this was observed in only 85%. Furthermore, in an era where we achieve a consistent vessel patency even in the more challenging



CTO lesions of 95%^{20,21,22,23}, TOSCA-2 patients had TIMI 3 flow only in 75% at the time of follow-up angiography. The above-mentioned failure to achieve a contemporary success rate in the TOSCA-2 lesions may impair the actual assessment of the benefit of the open artery hypothesis. In fact, the analysis of LV remodelling in patients with patent versus occluded infarct related lesions at follow-up, irrespective of the treatment arm, showed a highly significant beneficial effect on LV volume after one year with lower end-systolic and end-diastolic volumes⁸.

OAT and CTOs

While we have clear evidence for the benefit of acute PCI treatment for acute MI, OAT shows that delayed revascularisation of occluded infarct related arteries does not provide a prognostic benefit, at least in a patient subset that fulfils the clinical characteristics of the OAT population. We have to acknowledge that we have no randomised trial for even later intervention in occluded arteries, which are termed chronic total coronary occlusions (CTOs) if the occlusion duration is more than three months⁵. Should – or can – the findings of OAT as the single multicentre, randomised trial in sub-acutely occluded infarct-related arteries be extended to actual chronic occlusions?

The indication to reopen a CTO is a disputed subject in interventional cardiology. In the most recent PCI guidelines, patients with a CTO would represent typically patients with stable angina who would be eligible for a PCI according to the criteria for stable angina lesions. In addition, if the PCI of the CTO is considered to have a low success rate it is not advised to do perform PCI^{24,25}.

A recent review of the PCI experience at the Mayo Clinic over the past decades reveals that CTOs make up less than 5% of their PCI procedures^{26,27}. On the other hand, CTOs are found in a large fraction of patients with stable angina^{28,29}. In general, the major reason for this is that the PCI of a CTO is a complex, time and resources consuming procedure with a low procedural success rate, which requires, as well, more than average interventional skills and continuous and on-going personal experience of the operator^{30,31}. Randomised trials to show clinical benefit for the recanalisation of long-term occluded arteries are lacking, but the available nonrandomised long-term registries totalling more than 4,000 patients unanimously show even a benefit regarding the hard endpoint of survival, but are not unanimously accepted as clinical evidence^{32,33}. Still, registry data are accumulating, underscoring what sound clinical reasoning would suggest, e.g. that the presence of a CTO will put a patient at considerably higher risk if he experiences an acute MI in one of the remaining open arteries. Two independent studies showed a three-fold increase in mortality in patients with a CTO and a subsequent acute MI16,18. The recently published analysis from the New York State Survey showed that incomplete revascularisation by PCI leaving a CTO untreated led to higher mortality... even during a short follow-up of three years³⁴.

Are these data invalid now after OAT?

The answer must be... No.

First of all OAT did not include CTOs, and the given angiographic characteristics in OAT and TOSCA-2 are atypical for CTOs. Patients with a CTO entered into studies and registries have a prior STEMI

only in about 50% of the cases, less than half have severe LV dysfunction, and the majority have multivessel disease^{23,35-37}. One of the reasons why a prior MI is not mandatory for a CTO are well-developed collateral networks, which may have been present already at the time of an acute occlusion and helped to preserve LV function and viability³⁸. In OAT, almost all patients had either no visible collaterals or Rentrop grade 1 collaterals, whereas in CTOs 85% of collaterals are of Rentrop grade 3³⁹. In general, CTOs with no or very poor collateralisation (Rentrop 0 and 1) are not considered good candidates for a PCI attempt⁵.

Summary

The OAT trial and its angiographic substudy TOSCA-2, along with a number of published commentaries, represents examples of overinterpretation of clinical study results. A study that achieved no statistically significant result for any of the study endpoints can only claim to have proven that their null hypothesis cannot be rejected. The lack of power due to a reduction of patient numbers by one third, and an unexpected low event rate, makes it not unlikely that another trial would be able to disprove the null hypothesis. These statistical facts should be accepted by the authors and commentators. Instead, the inconclusive results were interpreted in such a way that they might apply to patient populations which had not been represented in the original trial (like CTOs without prior MI, or with viable myocardium), or for patients who could have been included, such as multivessel high risk patients, but had not been so by the investigators choice. Furthermore, major problems of the study are ignored, such as accepting TIMI 2 flow as a procedural success.

Conclusion

The OAT Trial is a valid addition to the data we have to answer the open artery hypothesis early after an acute MI, but only for those patients within the limits of its inclusion criteria. The study will find its way into future guidelines for interventional therapy, but caution needed not to misinterpret the study results. Interventional therapy needs to be based on evidence based medicine (EBM), which was defined by Dr. Sackett as a conscientious, explicit and judicious use of current best evidence in making decisions about the care of individual patients⁴⁰. The practice of EBM means integrating individual clinical expertise with the best available external clinical evidence from systematic research. The OAT trial does not add any valuable data in our decision to revascularise a CTO, and the concept of the open artery based on clinical pathophysiology and non-randomised data, as discussed above, still hold true for CTOs and determines good clinical practice.

References

1. Hochman JS, Lamas GA, Buller CE, Dzavik V, Reynolds HR, Abramsky SJ, Forman S, Ruzyllo W, Maggioni AP, White H, Sadowski Z, Carvalho AC, Rankin JM, Renkin JP, Steg PG, Mascette AM, Sopko G, Pfisterer ME, Leor J, Fridrich V, Mark DB, Knatterud GL; Occluded Artery Trial Investigators. Coronary intervention for persistent occlusion after myocardial infarction. *N Engl J Med* 2006;355:2395-407.

2. White HD, Braunwald E. Applying the open artery theory: use of predictive survival markers. *Eur Heart J* 1998;19:1132-9.



3. Hillis LD, Lange RA. Myocardial infarction and the open-artery hypothesis. *N Engl J Med* 2006;355:2475-7.

4. Armstrong PW. Ruminating on OAT: is the case closed? *Eur Heart J* 2007;28:531-2.

5. Di Mario C, Werner G, Sianos G, Galassi A, Büttner J, Dudek D, Chevalier B, Lefèvre T, Schofer J, Koolen J, Sievert H, Reimers B, Fajadet J, Colombo A, Gershlick A, Serruys PW, Reifart N for the EuroCTO Club . European perspective in the recanalisation of Chronic Total Occlusions (CTO): consensus document from the EuroCTO Club. *EuroInterv.* 2007;3:30-43.

6. Hochman JS, Lamas GA, Knatterud GL, Buller CE, Dzavik V, Mark DB, Reynolds HR, White HD; Occluded Artery Trial Research Group. Design and methodology of the Occluded Artery Trial (OAT). *Am Heart J* 2005;150:627-42.

7. Califf RM. The GUSTO trial and the open artery theory. *Eur Heart J* 1997;18 Suppl F:F2-10.

8. Dzavík V, Buller CE, Lamas GA, Rankin JM, Mancini GB, Cantor WJ, Carere RJ, Ross JR, Atchison D, Forman S, Thomas B, Buszman P, Vozzi C, Glanz A, Cohen EA, Meciar P, Devlin G, Mascette A, Sopko G, Knatterud GL, Hochman JS; TOSCA-2 Investigators. Randomized trial of percutaneous coronary intervention for subacute infarct-related coronary artery occlusion to achieve long-term patency and improve ventricular function: the Total Occlusion Study of Canada (TOSCA)-2 trial. *Circulation* 2006;114:2449-57.

 Boden WE, Eagle K, Granger CB. Reperfusion strategies in acute ST-segment elevation myocardial infarction: a comprehensive review of contemporary management options. J Am Coll Cardiol 2007;50:917-29.

10. Keeley EC, Hillis LD. Primary PCI for myocardial infarction with STsegment elevation. *N Engl J Med* 2007;356:47-54.

11. Fernandez-Avilés F, Alonso JJ, Castro-Beiras A, Vázquez N, Blanco J, Alonso-Briales J, López-Mesa J, Fernández-Vazquez F, Calvo I, Martínez-Elbal L, San Román JA, Ramos B; GRACIA (Grupo de Análisis de la Cardiopatía Isquémica Aguda) Group. Routine invasive strategy within 24 hours of thrombolysis versus ischaemia-guided conservative approach for acute myocardial infarction with ST-segment elevation (GRACIA-1): a randomised controlled trial. *Lancet* 2004;364:1045-53.

12. Scheller B, Hennen B, Hammer B, Walle J, Hofer C, Hilpert V, Winter H, Nickenig G, Böhm M; SIAM III Study Group. Beneficial effects of immediate stenting after thrombolysis in acute myocardial infarction. *J Am Coll Cardiol* 2003;42:634-41.

13. Di Mario C, Bolognese L, Maillard L, Dudek D, Gambarati G, Manari A, Guiducci V, Patrizi G, Rusconi LC, Piovaccari G, Hibon AR, Belpomme V, Indolfi C, Olivari Z, Steffenino G, Zmudka K, Airoldi F, Panzarasa R, Flather M, Steg PG. Combined Abciximab REteplase Stent Study in acute myocardial infarction (CARESS in AMI). *Am Heart J* 2004;148:378-85.

14. Le May MR, Wells GA, Labinaz M, Davies RF, Turek M, Leddy D, Maloney J, McKibbin T, Quinn B, Beanlands RS, Glover C, Marquis JF, O'Brien ER, Williams WL, Higginson LA. Combined angioplasty and pharmacological intervention versus thrombolysis alone in acute myocardial infarction (CAPITAL AMI study). *J Am Coll Cardiol* 2005;46:417-24.

15 Boden WE, O'Rourke RA, Teo KK, Hartigan PM, Maron DJ, Kostuk WJ, Knudtson M, Dada M, Casperson P, Harris CL, Chaitman BR, Shaw L, Gosselin G, Nawaz S, Title LM, Gau G, Blaustein AS, Booth DC, Bates ER, Spertus JA, Berman DS, Mancini GB, Weintraub WS; COURAGE Trial Research Group. Optimal medical therapy with or without PCI for stable coronary disease. *N Engl J Med* 2007;356:1503-16.

16 van der Schaaf RJ, Vis MM, Sjauw KD, Koch KT, Baan J Jr, Tijssen JG, de Winter RJ, Piek JJ, Henriques JP. Impact of Multivessel Coronary Disease on Long-Term Mortality in Patients With ST-Elevation Myocardial Infarction Is Due to the Presence of a Chronic Total Occlusion. *Am J Cardiol* 2006;98:1165-9.

17. Ross AM, Coyne KS, Moreyra E, Reiner JS, Greenhouse SW, Walker PL, Simoons ML, Draoui YC, Califf RM, Topol EJ, Van de Werf F, Lundergan CF. Extended mortality benefit of early postinfarction reperfusion. GUSTO-I Angiographic Investigators. Global Utilization of Streptokinase and Tissue Plasminogen Activator for Occluded Coronary Arteries Trial. *Circulation* 1998;97:1549-56.

18. Moreno R, Conde C, Perez-Vizcayno MJ, Villarreal S, Hernandez-Antolin R, Alfonso F, Bañuelos C, Angiolillo DJ, Escaned J, Fernandez-Ortiz A, Macaya C. Prognostic impact of a chronic occlusion in a noninfarct vessel in patients with acute myocardial infarction and multivessel disease undergoing primary percutaneous coronary intervention. *J Invasive Cardiol* 2006;18:16-9.

19. Montalescot G, Barragan P, Wittenberg O, Ecollan P, Elhadad S, Villain P, Boulenc JM, Morice MC, Maillard L, Pansiéri M, Choussat R, Pinton P; ADMIRAL Investigators. Abciximab before Direct Angioplasty and Stenting in Myocardial Infarction Regarding Acute and Long-Term Follow-up. Platelet glycoprotein IIb/IIIa inhibition with coronary stenting for acute myocardial infarction. *N Engl J Med* 2001;344:1895-903.

20. Colombo A, Chieffo A. Drug-eluting stent update 2007: part III: Technique and unapproved/unsettled indications (left main, bifurcations, chronic total occlusions, small vessels and long lesions, saphenous vein grafts, acute myocardial infarctions, and multivessel disease). *Circulation* 2007;116:1424-32.

21. Werner GS, Schwarz G, Prochnau D, Fritzenwanger M, Krack A, Betge S, Figulla HR. Paclitaxel-eluting stents for the treatment of chronic total coronary occlusions: A strategy of extensive lesion coverage with drug-eluting stents. *Catheter Cardiovasc Interv* 2006;67:1-9.

22. Suttorp MJ, Laarman GJ, Rahel BM, Kelder JC, Bosschaert MA, Kiemeneij F, Ten Berg JM, Bal ET, Rensing BJ, Eefting FD, Mast EG. Primary Stenting of Totally Occluded Native Coronary Arteries II (PRISON II): a randomized comparison of bare metal stent implantation with sirolimus-eluting stent implantation for the treatment of total coronary occlusions. *Circulation* 2006;114:921-8.

23. Werner GS, Krack A, Schwarz G, Prochnau D, Betge S, Figulla HR. Prevention of lesion recurrence in chronic total coronary occlusions by paclitaxel-eluting stents. *J Am Coll Cardiol* 2004;44:2301-6.

24. Smith SC Jr, Feldman TE, Hirshfeld JW Jr, Jacobs AK, Kern MJ, King SB 3rd, Morrison DA, O'Neill WW, Schaff HV, Whitlow PL, Williams DO, Antman EM, Smith SC Jr, Adams CD, Anderson JL, Faxon DP, Fuster V, Halperin JL, Hiratzka LF, Hunt SA, Jacobs AK, Nishimura R, Ornato JP, Page RL, Riegel B; American College of Cardiology/American Heart Association Task Force on Practice Guidelines; ACC/AHA/SCAI Writing Committee to Update the 2001 Guidelines for Percutaneous Coronary Intervention. ACC/AHA/SCAI 2005 guideline update for percutaneous coronary intervention: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (ACC/AHA/SCAI Writing Committee to Update the 2001 Guidelines for Percutaneous Coronary Intervention: a report of the American College of Cardiology/American Heart Association Task Force on Practice Guidelines (ACC/AHA/SCAI Writing Committee to Update the 2001 Guidelines for Percutaneous Coronary Intervention: J Am Coll Cardiol 2006;47:e1-121.

25. Silber S, Albertsson P, Avilés FF, Camici PG, Colombo A, Hamm C, Jørgensen E, Marco J, Nordrehaug JE, Ruzyllo W, Urban P, Stone GW, Wijns W; Task Force for Percutaneous Coronary Interventions of the European Society of Cardiology. Guidelines for percutaneous coronary Interventions. The Task Force for Percutaneous Coronary Interventions of the European Society of Cardiology. *Eur Heart J* 2005;26:804-47.

26. Singh M, Rihal CS, Gersh BJ, Lennon RJ, Prasad A, Sorajja P, Gullerud RE, Holmes DR Jr.Twenty-five-year trends in in-hospital and



long-term outcome after percutaneous coronary intervention: a singleinstitution experience. *Circulation* 2007;115:2835-41.

27. Prasad A, Rihal CS, Lennon RJ, Wiste HJ, Singh M, Holmes DR Jr. Trends in outcomes after percutaneous coronary intervention for chronic total occlusions: a 25-year experience from the Mayo Clinic. *J Am Coll Cardiol* 2007;49:1611-8.

28. Delacretaz E, Meier B. Therapeutic strategy with total coronary artery occlusions. *Am J Cardiol* 1997;79:185-7.

29. Christofferson RD, Lehmann KG, Martin GV, Every N, Caldwell JH, Kapadia SR. Effect of chronic total coronary occlusion on treatment strategy. *Am J Cardiol* 2005;95:1088-91.

30. Bell MR, Berger PB, Menke KK, Holmes DR Jr. Balloon angioplasty of chronic total coronary artery occlusions: what does it cost in radiation exposure, time, and materials? *Cathet Cardiovasc Diagn* 1992;25:10-5.

31. Stone GW, Colombo A, Teirstein PS, Moses JW, Leon MB, Reifart NJ, Mintz GS, Hoye A, Cox DA, Baim DS, Strauss BH, Selmon M, Moussa I, Suzuki T, Tamai H, Katoh O, Mitsudo K, Grube E, Cannon LA, Kandzari DE, Reisman M, Schwartz RS, Bailey S, Dangas G, Mehran R, Abizaid A, Serruys PW. Percutaneous recanalization of chronically occluded coronary arteries: procedural techniques, devices, and results. *Catheter Cardiovasc Interv* 2005;66:217-36.

32. Suero JA, Marso SP, Jones PG, Laster SB, Huber KC, Giorgi LV, Johnson WL, Rutherford BD. Procedural outcomes and long-term survival among patients undergoing percutaneous coronary intervention of a chronic total occlusion in native coronary arteries: a 20-year experience. *J Am Coll Cardiol* 2001;38:409-14.

33. Hoye A, van Domburg RT, Sonnenschein K, Serruys PW. Percutaneous coronary intervention for chronic total occlusions:

the Thoraxcenter experience 1992-2002. *Eur Heart J* 2005; 26:2630-6.

34. Hannan EL, Racz M, Holmes DR, King SB 3rd, Walford G, Ambrose JA, Sharma S, Katz S, Clark LT, Jones RH. Impact of completeness of percutaneous coronary intervention revascularization on long-term outcomes in the stent era. *Circulation* 2006;113:2406-12.

35. Nakamura S, Muthusamy TS, Bae JH, Cahyadi YH, Udayachalerm W, Tresukosol D. Impact of sirolimus-eluting stent on the outcome of patients with chronic total occlusions. *Am J Cardiol* 2005;95:161-6.

36. Ge L, lakovou I, Cosgrave J, Chieffo A, Montorfano M, Michev I, Airoldi F, Carlino M, Melzi G, Sangiorgi GM, Corvaja N, Colombo A. Immediate and mid-term outcomes of sirolimus-eluting stent implantation for chronic total occlusions. *Eur Heart J* 2005;26:1056-62.

37. Hoye A, Tanabe K, Lemos PA, Aoki J, Saia F, Arampatzis C, Degertekin M, Hofma SH, Sianos G, McFadden E, van der Giessen WJ, Smits PC, de Feyter PJ, van Domburg RT, Serruys PW. Significant reduction in restenosis after the use of sirolimus-eluting stents in the treatment of chronic total occlusions. *J Am Coll Cardiol* 2004;43:1954-8.

38. Werner GS, Surber R, Kuethe F, Emig U, Schwarz G, Bahrmann P, Figulla HR. Collaterals and the recovery of left ventricular function after recanalization of a chronic total coronary occlusion. *Am Heart J* 2005;149:129-37.

39. Werner GS, Ferrari M, Heinke S, Kuethe F, Surber R, Richartz BM, Figulla HR. Angiographic assessment of collateral connections in comparison with invasively determined collateral function in chronic coronary occlusions. *Circulation* 2003;107:1972-7.

40. Sackett DL, Rosenberg WM. The need for evidence-based medicine. *J R Soc Med* 1995;88:620-4.

